Author's response to reviews

Title: A Case-Based Approach to Shared Decision Making Skills Training for Clinicians

Authors:

Robert J Volk (bvolk@mdanderson.org)
Navkiran K Shokar (navkiran.shokar@ttuhsc.edu)
Viola B Leal (vbenavidez@mdanderson.org)
Robert J Bulik (bob.bulik@gmail.com)
Suzanne K Linder (sulinder@utmb.edu)
Patricia Dolan Mullen (patricia.d.mullen@uth.tmc.edu)
Richard M Wexler (rwexler@healthwise.org)
Gurjeet S Shokar (Gurjeet.Shokar@ttuhsc.edu)

Version: 4 Date: 29 September 2014

Author's response to reviews:

Dear Editors,

Thank you for catching this oversight in the discussion section of our revised manuscript. We have rewritten the specific sentence as follows to reflect the study addressing at least 2 of Kirkpatrick’s levels:

"We found that the use of an online case-based approach to teaching SDM, grounded in a conceptual model of the SDM steps used in a clinical encounter, led to positive results for at least two of the four Kirkpatrick’s levels of learning."

The revised manuscript is attached. Again, we appreciate your consideration of our manuscript. Sincerely, Bob Volk.